Author's response to reviews

Title: Revision in total hip arthroplasty using a ceramic revision head: Two year results of eight patients

Authors:

Dirk Ganzer (ganzerdr@t-online.de)
Lutz Forke (forke@ms-arn.de)
Ulrich Irlenbusch (irlenbusch@ms-arn.de)

Version: 5 Date: 27 August 2014

Author's response to reviews: see over
Author's response to reviews

Title: Revision total hip arthroplasty using a ceramic revision head with a retained well-fixed femoral component: Two year results in eight patients

Authors:
Dirk Ganzer (ganzerdr@t-online.de)
Lutz Forke (forke@ms-arn.de)
Ulrich Irlenbusch (irlenbusch@ms-arn.de)

Version: 4 Date: 28 August 2014

Author's response to reviews: see over
Reviewer’s Report

Title: Two-year follow-up of revision total hip arthroplasty using a ceramic revision head with a retained well-fixed femoral component: a case series of eight patients

Version: 4 Date: 28 August 2014

Reviewer: Editor – Anna Kathrina Vidal

Dear Dr. Ganzer,

Thank you for revising your manuscript.

Unfortunately, not all the requested changes have been made, please address the following points:

1. Please confirm if you would like to change the article type to Case Report.
   
   • The authors confirm that we would like to change the article type to Case Report (in this instance, a Case Series). The manuscript has been presented as a case series according to the journal template and the guidance provided below.

2. Please also provide a point-by-point response to the reviewers’ reports, detailing where and how you have revised your manuscript. For an example of how we would like you to format your cover letter, please see the links below.

   http://www.biomedcentral.com/imedia/2041411268621582_comment.pdf
   
   http://www.biomedcentral.com/imedia/1865654785489471_comment.pdf

   • We hope that the response provided here meets the journal expectations.

3. Please enumerate the cases in the Case presentation section accordingly (e.g. Case 1 patient...Case 2 patient...)

   • The Case Presentation section has been extensively revised to fulfill these criteria. Please see the additional text from page 6 line 1 through page 8 line 4.

4. Please include the Ethnicity of the patient in the Abstract and Case presentation sections.

   • This information has now been included; please see Abstract page 2 line 11, and Case Presentation section page 6 line 1, page 6 line 9, page 6 line 16, page 6 line 22, page 7 line 3, page 7 line 10, page 7 line 17, and page 7 line 24.

5. As this is a case series, please confirm if all the patients gave their consent for the publication of this manuscript. We recommend the following wording is used for the consent section to be placed after the Conclusions section:

   "Written informed consent was obtained from the patients for publication of this case report and any accompanying images. A copy of the written consent is available for review by the Editor-in-Chief of this journal."

   • The authors confirm consent from all patients. Please see the text suggested above on page 14 of the manuscript.

We are very keen to continue with the peer-review process on your article; however, we cannot proceed until it is in the correct format.
• The authors hope that the revisions outlined above address all issues related to the format of the manuscript.

The revised version of the manuscript can be submitted by logging onto the submission system via the following link:

Thank you and I look forward to receiving the revised manuscript by 28 August 2014.

Best wishes,

Kathrina

Ms. Anna Kathrina Vidal
Journal Editorial Office
BioMed Central